Author’s response to reviews

Title: Vaginal douching in Zambia: A risk or benefit to women in the fight against cervical cancer: A retrospective cohort stud

Authors:

Twaambo Hamoonga (tehams24@gmail.com)
Pawel Olowski (pawelolowski@hotmail.com)
Patrick Musonda (pmuzho@hotmail.com)

Version: 1 Date: 02 Mar 2019

Author’s response to reviews:

The authors have revised the discussion section of the study. The authors have avoided using the direct words used from previous publications cited in the discussion.